Author’s response to reviews

Title: Mapping infectious disease hospital surge threats to lessons learnt in Singapore: A systems analysis and development of a framework to inform how to DECIDE on planning and response strategies

Authors:

Shweta Singh (shweta@nuhs.edu.sg)
Richard Coker (richard.coker@lshtm.ac.uk)
Hubertus Vrijhoef (vrijhoef_hubertus@nuhs.edu.sg)
Yee Sin Leo (yee_sin_leo@ttsh.com.sg)
Angela Chow (angela_chow@ttsh.com.sg)
Mark I-Cheng Chen (mark_chen@nuhs.edu.sg)
Poh Lian Lim (poh_lian_lim@ttsh.com.sg)
Qinghui Tan (tan_qing_hui@brc.a-star.edu.sg)
Zoe Hildon (Zhildon1@jhu.edu)

Version: 2 Date: 14 Aug 2017

Author’s response to reviews:

We are glad both reviewers found our revisions acceptable, and grateful for all their useful feedback.

We upload the new documents as follows:

1. Including all the sections as listed below -
   Abbreviations
Ethics approval and consent to participate
Consent for publication
Availability of data and material
Competing interests

Funding

Authors' contributions

Acknowledgements

2. Providing figure titles/legends under a separate heading of 'Figure Legends' after the References. We removed the Figure titles/legends from within the main text of the manuscript.

3. Figure files now only contain the image/graphic, as well as any associated keys/annotations.

4. The previously named "Appendix I" is renamed as a supplementary file in the text and added as such, removing the footnote “Appendix I”.

5. Uploading our manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files are also clean versions. Figures (and additional files) remain uploaded as separate files.

Please note we uploaded one figure in colour; if you require a non-colour version please do let us know, although the colour version is preferable if this is acceptable.

Many thanks for all you help and support with this publication, if you need any further adjustments please do not hesitate to contact us.

Kind regards,

Zoe